Author’s response to reviews

Title: Validating an alternate version of the chewing function questionnaire in partially dentate patients

Authors:

kazuyoshi Baba (kazu@senzoku.showa-u.ac.jp)
Mike T John (johnx055@umn.edu)
Mika Inukai (mika-i.rpro@tmd.ac.jp)
Kumiko Aridome (aripro@tmd.ac.jp)
Yoshimasa Igarashi (igarashi.rpro@tmd.ac.jp)

Version: 2 Date: 24 December 2008

Author’s response to reviews:

December 24, 2008

Dear Dr. Rikki Graham, PhD
Senior Assistant Editor
BMC-series journals

We are submitting our revision of the manuscript titled “Validating an alternate version of the chewing function questionnaire in partially dentate patients” We believe we have responded to the reviewers’ comments and have incorporated the suggested changes into the revised manuscript. We have provided point by point responses to the reviewers’ comments and information on how we changed the manuscript. We would be very pleased if the manuscript is judged acceptable for publication in the BMC Oral Health.

Very sincerely yours,

Kazuyoshi Baba, D.D.S., Ph.D.
Professor and chair
Department of Prosthodontics, Showa University
Email: kazu@senzoku.showa-u.ac.jp